**Author’s response to reviews**

**Title:** Transcultural adaption and preliminary evaluation of “Understanding low back pain” patient education booklet

**Authors:**
Anna-Sofia Simula (anna.simula@oulu.fi)
Hazel Jenkins (hazel.jenkins@mq.edu.au)
Riikka Holopainen (riikka.holopainen@wippies.fi)
Petteri Oura (petteri.oura@oulu.fi)
Katariina Korniloff (katariina.korniloff@jamk.fi)
Arja Häkkinen (arja.h.hakkinen@jyu.fi)
Esa-Pekka Takala (Esa-Pekka.Takala@ttl.fi)
Mark Hancock (mark.hancock@mq.edu.au)
Jaro Karppinen (jaro.karppinen@oulu.fi)

**Version: 2 Date: 16 Dec 2019**

**Author’s response to reviews:**

Dear Editor,

Please find enclosed our revised manuscript entitled Transcultural adaption and preliminary evaluation of “Understanding low back pain” patient education booklet by Anna-Sofia Simula, Hazel J Jenkins, Riikka Holopainen, Petteri Oura, Katariina Korniloff, Arja Häkkinen, Esa-Pekka Takala, Mark J. Hancock and Jaro Karppinen.

We have carefully responded to the comments of the editor. Thus, we would be grateful if our manuscript could be re-evaluated for publication in BMC Health Services Research.

1. We have made clarifications to the funding section.

2. We have uploaded a clean version of the manuscript.

Sincerely, on behalf of the authors,
Anna-Sofia Simula

Center for Life Course Health Research, Faculty of Medicine, University of Oulu, Oulu, Finland
Email annasofia.simula@gmail.com
Tel +358445758125
PO Box 5000, FI-90014 University of Oulu, Finland

December 16th, 2019

Oulu, Finland